Author’s response to reviews

Title: Clinical, biochemical and metabolic characterization of patients with short-chain enoyl-CoA hydratase (ECHS1) deficiency: two case reports and the review of the literature

Authors:

Hua Yang (397158301@qq.com)

Dan Yu (yd540@126.com)

Version: 1 Date: 04 Jan 2020

Author’s response to reviews:

Authors' Response to Reviewers' Comments

Number : BPED-D-19-00886

Manuscript title: Clinical, biochemical and metabolic characterization of two cases with short-chain enoyl-CoA hydratase (ECHS1) deficiency

Dear Editor and reviewers,

Thank you for your letter, regarding the above manuscript. We have analysed the comments carefully and we would like to thank you for your time reviewing our manuscript. We have attempted to address each issue the reviewers have raised regarding our paper. As a result, we revised our manuscript seriously according to the comments, and we hope that the reviewers and you will assess its acceptability for publication.

Best regards

(1) "The background and case presentation section include repeats (i.e. treatment…)."

Authors’ response: Repetitions made in the background and case presentation have been reviewed and necessary amendments have been made.

(line 7 in case presentation part)
(2) 'The review of the literature' may be added in title."

Author’s response: The title of the article has been revised accordingly.

(3) "Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format."

Authors’ response: I have re-uploaded the report to meet the requirements.

(4) "If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English."

Authors’ response: My manuscript has been reviewed by Dr. Mowshica Rajah.

(5) "In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Pediatrics must contain a Declarations section which includes the mandatory sub-sections listed below."

revised part: my manuscript submission contains the following: Declarations: Ethics approval and consent to participate, Consent to publish, Availability of data and materials, Competing interests, Funding, Authors’ Contributions, Acknowledgements. Furthermore, in the Acknowledgement I have included "and also thank Dr. Mowshica Rajah for reviewing the manuscript.".